## **Response to Reviewers Comments for SDC101**

We have made minor addition to pg 6, 7, stating the use of economic analysis to quantify the effect of BMPs, and have added a paragraph to pg 11, and 12 outlining the project going beyond just as the research, but extensive outreach and extension activities.

We have added two Economists to the project. Dr. Pengfei Liu (University of Rhode Island) and Dr. Tatiana Borisova (University of Florida). I am also told that there may be additional Economist from TAMU who might be added to this project.